Author's response to reviews

Title: Do inter-hospital comparisons of in-hospital, acute myocardial infarction case-fatality rates serve the purpose of fostering quality improvement? An evaluative study.

Authors:

Willem Aelvoet (willem.aelvoet@health.fgov.be)
Nathalie Terryn (nathalie.terryn@health.fgov.be)
Geert Molenberghs (geert.molenberghs@uhasselt.be)
Guy De Backer (guy.debacker@ugent.be)
Christiaan Vrints (Chris.vrints@uza.be)
Marc van Sprundel (marc.vansprundel@ua.ac.be)

Version: 4 Date: 6 December 2010

Author’s response to reviews: see over
Dear editor,

Herewith we submit the revised manuscript entitled “Do inter-hospital comparisons of in-hospital, acute myocardial infarction case-fatality rates serve the purpose of fostering quality improvement? An evaluative study.”

Table 2 has now been included in the manuscript
Line numbers as well as highlighting/tracking have been removed.